Author's response to reviews

Title: Absent otoacoustic emissions predict otitis media in young Aboriginal children: a birth cohort study in Aboriginal and non-Aboriginal children in an arid zone of Western Australia

Authors:

Deborah Lehmann (deborahl@ichr.uwa.edu.au)
Sharon Weeks (mbra6582@bigpond.net.au)
Peter Jacoby (peterj@ichr.uwa.edu.au)
Dimity Elsbury (dimityelsbury@yahoo.com.au)
Janine Finucane (janine.finucane@health.wa.gov.au)
Annette Stokes (annette_stokes@hotmail.com)
Ruth Monck (clrmonck@emerge.net.au)
Harvey Coates (harveyc@cyllene.uwa.edu.au)
Kalgoorlie Otitis Media Research Project Team (KOMRP) (deborahl@ichr.uwa.edu.au)

Version: 8 Date: 23 August 2008

Author's response to reviews:

Author group - name included within list of authors and details in acknowledgements. Please note that I have put my email address against this group name and leave you to edit accordingly - submission is not possible without an email. Thank you.

Figure 3 - shading removed. There is no shading in tables.